Author’s response to reviews

Title: Seroprevalence and associated factors of HSV-2 infection among general population in Shandong Province, China

Authors:

Pengcheng Huai (huaipengcheng@163.com)
Furong Li (lfr2008@foxmail.com)
Zhen Li (lizhen19871012@163.com)
Lele Sun (438065194@qq.com)
Xi’an Fu (f19830827@126.com)
Qing Pan (349385789@qq.com)
Gongqi Yu (yugongitje@163.com)
Zemin Chai (359342307@qq.com)
Tongsheng Chu (tongshengchu@163.com)
Zihao Mi (mizihao2015@126.com)
Fangfang Bao (baofangfang2008@163.com)
Honglei Wang (hongleiwang86@163.com)
Bingni Zhou (383118538@qq.com)
Chuan Wang (wangchuan86@126.com)
Yonghu Sun (suohandong@126.com)
Guiye Niu (miss2001flyn@126.com)
Yuan Zhang (zhangyuanbj@sina.cn)
Fanghui Fu (1259934806@qq.com)
Xiaoqiao Lang (289308838@qq.com)
Xiaoling Wang (472634899@qq.com)
Hui Zhao (zhaohui7867@qq.com)
Daina Liu (1558179042@qq.com)
Hong Liu (hongyue2519@hotmail.com)
Dianchang Liu (dianchangliu@163.com)
Jian Liu (liujian751123@163.com)
Aiqiang Xu (aqxuepi@163.com)
Furen Zhang (zhangfuren@hotmail.com)

**Version: 2 Date: 03 Apr 2019**

**Author’s response to reviews:**

Dear Prof. Oliver Cocks,

Thank you very much for you to give us an opportunity to do minor revision. We have addressed all the comments raised by editor. All revisions in the manuscript are described in this response letter. Please find our point-to-point reply to the comments below.

Please feel free to contact us if you have any question about the revision and we are looking forward to your good news.

Best regards,

Very Respectfully,

Pengcheng Huai
Title: Seroprevalence and risk factors of HSV-2 infection among general population in Shandong Province, China: a cross-sectional study

Authors: Pengcheng Huai; Furong Li; Zhen Li; Lele Sun; Xi’an Fu; Qing Pan; Gongqi Yu; Zemin Chai; Tongsheng Chu; Zihao Mi; Fangfang Bao; Honglei Wang; Bingni Zhou; Chuan Wang; Yonghu Sun; Guiye Niu; Yuan Zhang; Fanghui Fu; Xiaoqiao Lang; Xiaoling Wang; Hui Zhao; Daina Liu; Hong Liu; Dianchang Liu; Jian Liu; Aiqiang Xu; Furen Zhang

Submitted Journal: BMC Infectious Diseases

Editor Comments:

1. Please format your abstract under the following headings:
   • Background
   • Methods
   • Results
   • Conclusions
   • Keywords

More information on what should be included in these sections can be found at https://bmcinfectdis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Response: Thank you very much for your comment. The abstract has been revised according to the above requirement.
2. Please change the section heading “Acronyms” to “Abbreviations”.

Response: Thank you for your comment. “Acronyms” has been replaced by “Abbreviations”.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Thank you very much for your comment. The manuscript, tables and figures have been revised according to the above requirement.

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Response: Thank you very much for your suggestions. The language of our manuscript was reviewed by International Science Editing.

------------------

Editorial Policies
------------------

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements, this may cause a delay while this is addressed. Failure to adhere to our policies may result in rejection of your manuscript.
In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Infectious Diseases must contain a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcinfectdis.biomedcentral.com/).

Where a mandatory Declarations section is not relevant to your study design or article type, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).

Declarations

-Ethics approval and consent to participate

-Consent to publish

-Availability of data and materials

-Competing interests

-Funding

-Authors' Contributions

-Acknowledgements

Response: Thank you very much for your comments. This manuscript adhere to all the above editorial policies and requirements.